Author's response to reviews

**Title:** The Relationship Between Microvascular Complications And Vitamin D Deficiency In Type-2 Diabetes Mellitus

**Authors:**

Celil Usluogullari (calperuslu@yahoo.com)
Fevzi Balkan (fbalkan21@gmail.com)
Sedat Caner (drsedat@yahoo.com)
rifki ucler (rifkiucler@gmail.com)
Cafer Kaya (caferk@gmail.com)
Reyhan Ersoy (drreyhan@yahoo.com)
Bekir Cakir (drbekir@gmail.com)

**Version:** 4  **Date:** 23 September 2014

**Author's response to reviews:**

Answer sto editor
Dear editor
We tried to make your advise.
Congurlations.

(1) The authors need to describe the study population in more detail in order for the reader to be able to understand the results in more depth. In particular, there need to be comparisons between those with vit D deficiency vs. normal vitamin D and microvascular complications or nephropathy vs. those without. Depending on the findings of such comparisons further multivariable adjustments might need to be conducted to assess the relationship between vit D and the study outcomes. Also it is important to provide a subgroup analysis by gender.

We corrected your advise in manuscript.(red line)